Author's response to reviews

Title: Methodological Aspects of a GIS-based Environmental Health Inspection Program used in the Athens 2004 Olympic and Para Olympic Games

Authors:

Christos Hadjichristodoulou (hadjich@ath.forthnet.gr)
Elpidoforos S Soteriades (esoteria@hsph.harvard.edu)
Virginia Kolonia (hadjich@ath.forthnet.gr)
Matthew E Falagas (matthew.falagas@tufts.edu)
Efstathios Pantelopoulos (hadjich@ath.forthnet.gr)
Georgios Panagakos (hadjich@ath.forthnet.gr)
Varvara Mouchtouri (hadjich@ath.forthnet.gr)
Jeni Kremastinou (hadjich@ath.forthnet.gr)

Version: 2 Date: 21 March 2005

Author's response to reviews: see over
Athens, March 10, 2005

To: Editor,  
   BMC Public Health

Re: Paper submission

Dear Editor,

We are pleased to submit our manuscript: “Methodological Aspects of a GIS-based Environmental Health Inspection Program used in the Athens 2004 Olympic and Paralympic Games” for consideration in the BMC Public Health.

We believe that our experience in the development and implementation of an environmental health inspection program during the Athens 2004 Olympic games using geographic information systems may benefit other public health professionals in similar endeavors.

The manuscript has not been submitted elsewhere, and if accepted in the BMC Public Health, will not be republished in any other journal in the same or similar form without the written consent of the Editor of the BMC Public Health. There are no financial or other relationships that might lead to a conflict of interest. The manuscript has been read and approved by all authors.

If you have any questions or need additional information please feel free to contact me at +30 (210) 693 226 4685 or via e-mail at the given address. I would like to thank you in advance for your consideration.

Sincerely,

Dr. Christos Hadjichristodoulou,  
Associate professor,  
Department of Hygiene and Epidemiology  
University of Thessaly School of Medicine  
22 Papakyriazi Str.,  
41222 Larissa, Greece  
Tel: +30 2410 565259, +30 6932264685,  
Fax: +30 210 5311040,  
E-mail: hadjich@ath.forthnet.gr